Author's response to reviews

Title: A Rare Cause Of Abnormal Uterine Bleeding

Authors:

Mridula A Benjamin (mridulaben@yahoo.com)
Hjh Roselina Yaakub (yaakubroselina@hotmail.com)
PU Telesinghe (yaakubroselina@hotmail.com)
Gazala Kafeel (yaakubroselina@hotmail.com)
Miriam Kingsley (Miriam.kingsley@biomedcentral.com)

Version: 2 Date: 12 January 2010

Author's response to reviews:

From
Dr. Mridula A. Benjamin
Dept of Obs and Gyn
RIPAS Hospital
Bandar Seri Begawan
Brunei

To
Journal of Medical Case Reports Editorial Team
Cases Network Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK
T: 020 7631 9930

Dear Sir/madam,

I had submitted an article for Journal of Medical Case Reports titled “A RARE CAUSE OF ABNORMAL UTERINE BLEEDING” (Submission no 6958). This has been peer reviewed and a few changes have been suggested. As requested I have made the following changes.

1. POD (abbreviation meaning): Changed to “Pouch of Douglas”

2. INR of the patient was 2, which is abnormal, but no comment is made as to the reason for this: It is mentioned that there was a suspicion of coagulation disorder while other causes were being ruled out.

3. ’a possible spurter from the cervix’: Changed to “active bleeding point from the cervix”.
4. operating 'theatre': Changed to “operating room”.
5. the possibility of 'ramifying haemangiomas': Changed to “the differential diagnosis of ramifying haemangiomas”
6. References have been arranged as advised.
If any other changes need to be made, I may be kindly informed.

With regards,
Dr. Mridula A Benjamin